

NOTA DI LAVORO 48.2015

Optimal Management of Markets for Bankable Emission Permits

Jussi Lintunen, Natural Resources Institute Finland Olli-Pekka Kuusela, Natural Resources Institute Finland

Climate Change and Sustainable Development Series Editor: Carlo Carraro

Optimal Management of Markets for Bankable Emission Permits

By Jussi Lintunen, Natural Resources Institute Finland (Luke) Olli-Pekka Kuusela, Natural Resources Institute Finland (Luke)

Summary

We examine the optimal management of emission permit markets when banking but not borrowing of permits is allowed. The regulator maximizes expected social welfare through an optimal allocation rule in an infinite horizon setting. The policy is second-best as the emission cap is set before the uncertainty about the current state of the economy is resolved. In this setting, the role of banking is to decrease the regulator's risk as it generates an endogenous price floor in the permit markets. We show that the regulator's optimal policy adjusts the emissions cap irrespective of the existing number of permits in the bank, with the implication that the regulator neutralizes the effect of the existing bank on future permit prices. We derive the optimality conditions for the second-best emission cap with banking and solve the model analytically in the case of IID shocks. Our results show that the discount factor together with the slopes of the marginal damages and benefits determine the welfare gains from allowing banking of permits. Finally, to address the current state of the EU Emission Trading Scheme (EU ETS) and guide the design of future permit markets, we solve the model numerically with persistent shock process and show that the optimal emission cap is positively correlated with business cycles, meaning that during downturns the regulator should tighten the cap. The expected emissions and permit prices also correlate positively with economic activity

Keywords: Cap and Trade, Climate Change, Business Cycle, Second Best, Prices vs. Quantities JEL Classification: E32, Q54, Q58

Address for correspondence Jussi Lintunen Natural Resources Institute Finland (Luke) Box 18, FI-01301 Vantaa Finland E-mail: jussi.lintunen@luke.fi

Optimal Management of Markets for Bankable Emission Permits

Jussi Lintunen and Olli-Pekka Kuusela^{*}

March 3, 2015

Abstract

We examine the optimal management of emission permit markets when banking but not borrowing of permits is allowed. The regulator maximizes expected social welfare through an optimal allocation rule in an infinite horizon setting. The policy is second-best as the emission cap is set before the uncertainty about the current state of the economy is resolved. In this setting, the role of banking is to decrease the regulator's risk as it generates an endogenous price floor in the permit markets. We show that the regulator's optimal policy adjusts the emissions cap irrespective of the existing number of permits in the bank, with the implication that the regulator neutralizes the effect of the existing bank on future permit prices. We derive the optimality conditions for the second-best emission cap with banking and solve the model analytically in the case of IID shocks. Our results show that the discount factor together with the slopes of the marginal damages and benefits determine the welfare gains from allowing banking of permits. Finally, to address the current state of the EU Emission Trading Scheme (EU ETS) and guide the design of future permit markets, we solve the model numerically with persistent shock process and show that the optimal emission cap is positively correlated with business cycles, meaning that during downturns the regulator should tighten the cap. The expected emissions and permit prices also correlate positively with economic activity.

JEL classification: E32, Q54, Q58

Keywords: cap and trade, climate change, business cycle, second best, prices vs. quantities

^{*}Natural Resources Institute Finland (Luke). J. Lintunen: Box 18, FI-01301, Vantaa, Finland. jussi.lintunen@luke.fi, O.-P. Kuusela: olli-pekka.kuusela@luke.fi

1 Introduction

Possibility to bank permits is a common feature in many of the current and proposed emission permit systems such as The European Union Emissions Trading Scheme (EU ETS) and The American Power Act.¹ The motivation for including banking provisions is commonly framed as a mean to provide intertemporal flexibility and improved efficiency in abatement effort as well as protection against random fluctuations in economic environment (Chevallier, 2012). Recent experiences in EU ETS have, however, led to demands for market interventions such as "backloading" to increase the prevailing permit prices viewed as too low to be effective and also to moderate the quantity of permits held in the bank.² The need for such arbitrary changes raises concerns over the future viability of permit schemes, and going forward, it is likely that rule-based interventions will be preferred to regulator's discretion. Surprisingly, the current literature on permit markets with banking does not provide such rules that aim at maximizing social welfare.³

The purpose of our paper is to examine the optimal and active management of permit markets in the presence of uncertainty and the possibility to bank permits for future use. Our analysis contributes to the past research on the second-best policy design by explicitly introducing a regulator whose goal is to maximize social welfare through an optimal allocation rule in an infinite horizon setting. Like in Schennach (2000), our main focus is on such permit systems where banking is allowed but borrowing is not.⁴ The need for adjusting the periodic allocation stems from the fact that the regulator sets the emissions cap before learning the current state of the economy, whereas firms and speculators have the advantage of observing the current state prior to their actions. Given this information structure, our work also continues in the spirit of Weitzman's (1974) seminal paper by determining when a permit system with banking dominates tax-based regulations and permit systems without banking in welfare terms.

The literature on permit banking identifies two main motives for allowing permit banking in cap and trade schemes. The first motive is deterministic and driven by discounting, whereas the second one is framed in terms of inherent uncertainty in the ultimate cost of abatement (Fell et al., 2012). Deterministic banking motive refers to firms' decision to smooth abatement costs through time given that the periodic permit allocation does not satisfy inter-temporal optimality conditions (Cronshaw

¹The first successful emissions trading scheme was The Acid Rain Program established 1990 with the purpose of cutting the SO_2 emissions from electricity generation. It also allowed banking of unused permits for future use (Schennach, 2000).

²The EU Commission has put forth a policy proposal to create a reserve mechanism to adjust permit supply in times of unexpected demand shocks that threaten to drive the price too low and hence create a glut of banked permits.

 $^{^{3}}$ Newell et al. (2005) examine "quantity-plus policies" that aim at stabilizing the permit price at some desired level but their focus is on cost efficiency.

⁴Almost all of the current and proposed permit schemes considerably restrict borrowing from future allocations. For example, EU ETS implicitly allows borrowing from the next year's allocation, but the extent of such borrowing remains limited (Chevallier, 2012). Some recent papers have examined the welfare effect of various restrictions on borrowing (Fell et al., 2012; Leard, 2013)

and Kruse, 1996; Rubin, 1996). This smoothing may or may not improve welfare depending on the type of damages. Uncertainty banking, on the other hand, refers to firms' decision to hedge against future cost shocks as well as to smoothing of the effects of random fluctuations in input prices, demand and weather (Schennach, 2000).

We propose that banking also has a third motive that arises from the planner's active policy to maximize welfare and from the firms' speculative demand to bank permits. We show that the planner takes advantage of an endogenous price floor created by the expectations for future permit price. The expected price is in turn determined by the planner's policy and by any information conveyed in the most recent shock realization. This third motivation for banking has not been explicitly stated in the literature and its presence provides some new results. In effect, the presence of an endogenous price floor creates a hybrid policy that combines the properties of the quantity and price policies.⁵ The planner can then utilize the presence of an endogenous price floor by increasing the level of the optimal cap relative to no-banking case. The reason for this is that the presence of a price floor reduces the efficiency losses incurred at low shock levels. This in turn enables the planner to hedge against efficiency losses at high shock levels by relaxing the cap.

Kling and Rubin (1997) were first to study social efficiency of inter-temporal permit trading, with both banking and borrowing provisions. Their analysis compares emissions in an unregulated market outcome to the socially optimal level of emissions, and they propose an inter-temporal trading ratio (ITR) that can achieve the latter. They conclude that without ITR, inter-temporal trading may not be welfare improving. Yates and Cronshaw (2001) introduce uncertainty in the problem of second-best policy. They use a two-period model to examine whether inter-temporal trading improves efficiency and if so, how the system should be optimally designed using ITRs. They find that inter-temporal trading should be allowed only when the slope parameter of marginal abatement costs is greater than that of the marginal damage function. They also provide a preliminary discussion of a full-fledged dynamic setting in which the regulator can update the allocation policy. Our paper is most closely related to paper by Feng and Zhao (2006) which refines the analysis in Yates and Cronshaw (2001) by distinguishing between three separate effects of banking on welfare. Their focus is on comparing two different ITRs: unitary and monetary equivalent interest rate. We show that in a dynamic framework where borrowing is restricted some of their results do not continue to hold. For example, in most relevant cases, the cap is set at a higher level than in permit systems without banking. We also show that the optimal number of new permits allocated completely neutralizes the effect of the existing bank of permits.

The previous literature has mostly concentrated on emission trading schemes where the regulator sets the allocation path of new permits at the time of introduction of the policy. Newell et al. (2005) is

 $^{{}^{5}}$ Roberts and Spence (1976) were first to propose a hybrid policy with price floors and ceilings. Williams (2002) discusses the need for research for such policies in combination with banking. Newell and Pizer (2003) mention that inter-temporal trading of permits makes quantity policies behave more like a price policy.

an exception. They examine whether banking provisions can be used to emulate the properties of tax based instruments in a setting where marginal damages are relatively flat. They hypothesize that the regulator may want to achieve a certain expected permit price level through active management of the policy through periodic allocation rules. They, however, allow for borrowing of permits which alters the permit markets considerably. Most recently, Fell et al. (2012) and Leard (2013) examine welfare properties of permit banking in a multi-period setting, but they do not solve for the optimal permit allocation rule under uncertainty and they do not allow for active management of the cap levels. Defining such an optimal rule is one of the main contributions of our paper.

We first analyze the planner's optimal policy in a special case where the periodic random shocks are independent and identically distributed. As to be expected, we find that the policy performs better the closer it can follow the marginal damage curve. In the case where marginal damages are flat, like with greenhouse gas (GHG) emissions, banking allows the planner's policy to approach the marginal damage curve. This occurs especially when the discount factor is close to unity, or alternatively, when the length of time between allocations is short. However, with steep marginal damages, banking seems to distort the policy away from the marginal damages. In an intermittent case banking may yield welfare gains compared to the quantity policy without banking or to a tax policy. Finally, we use our model to examine the optimal management of present-day cap and trade systems such as EU ETS. We show that in the presence of persistent business cycles, the planner acting optimally makes considerable adjustments to the periodic cap while keeping the expected permit price at a constant level. During economic downturns, the cap is tightened, and during upturns, it is relaxed. Adjustments are done in such a manner that during periods of high economic activity, the expected permit price is higher than in periods of low activity. We conclude that with persistent business cycles there is an even greater need for optimal allocation rules for new permits.

The rest of the paper is organized as follows. Section 2 presents the description of the economy and the permit market outcome. Section 3 presents the planner's problem and the resulting optimal policy. Section 4 describes the optimal policy under both IID and persistent shocks as well as in the case of climate change mitigation policy. Section 5 concludes.

2 The economy and permit markets

Following the setup in Feng and Zhao (2006), the economy produces a consumption good, q_t , that yields stochastically evolving net benefits (surpluses) given by a function $B(q_t, \theta_t)$.⁶ The marginal benefit is strictly decreasing, $B_{qq}(q, \theta) < 0$, and satisfies conditions $B_q(0, \theta) > 0$ and $\lim_{q\to\infty} B_q(q, \theta) < 0$.⁷ Random variable, θ_t , represents market fluctuations such as shifts in consumer demand and cost of

 $^{^{6}}$ Other interpretations of q are also possible, for example, it can represent the use of some unspecified input.

⁷Notation B_{qq} has the standard interpretation as the second derivative of benefit function with respect to q. The variables are, in general, time dependent. To avoid notational clutter, we omit the time indices whenever there is no risk of confusion.

production, with high realizations leading to higher marginal net benefits from q, $B_{q\theta} > 0.^8$ After the current value of θ_t is realized, the optimization problem $\max_q B(q, \theta)$ determines the unregulated outcome, and the unregulated level of production is implicitly defined by the problem's first order condition:

$$B_q(q,\theta) = 0. \tag{1}$$

Denote the unregulated solution by $q_0(\theta)$. By the implicit function theorem, $q'_0(\theta) > 0$, that is, the quantity produced and consumed is increasing in θ .

Production of q_t generates environmentally harmful emissions, treated here as an externality. We define the resulting environmental damages in the next section when we present the planner's problem. To simplify notation, we measure q_t in units of pollution emitted. A regulator, whose goal is to limit emissions from good q_t , imposes a unit price p_t on emissions. This price can be either a Pigouvian tax or alternatively, a price of an emission permit. The regulated outcome in both cases is determined by the following optimization problem:

$$\max B(q,\theta) - pq.$$

The regulated production level, $q = q(p, \theta)$, is implicitly given by the first order condition

$$B_q(q,\theta) - p = 0. \tag{2}$$

The following properties hold by the implicit function theorem: $q_p(p,\theta) = B_{qq}^{-1} < 0$ and $q_\theta(p,\theta) > 0$. That is, by increasing the unit price of emissions the regulator can reduce the quantity emitted. Note that unregulated outcome $q'_0(\theta) \equiv q_\theta(0,\theta)$.

Under a tax system the unit price of emissions is exogenously given. In a cap and trade system, however, the price of a permit is endogenously determined, and it depends on the number of permits available, Q_t , and the current shock term, θ_t . If the unregulated market emissions, $q_0(\theta)$, are less than the cap, the cap is not binding and the price of the permit will be zero, and when the emission cap is binding, there will be a positive permit price. These equilibrium conditions for the permit market can be written as

$$p \ge 0, \qquad Q - q \ge 0 \qquad \text{and} \qquad (Q - q)p = 0.$$
 (3)

With no inter-temporal trade of permits allowed, the regulator simply voids all excess permits at the end of each period. If banking of permits is allowed, then it becomes possible to carry over current permits to the next period.⁹ To model this, we introduce speculative banking demand for permits, denoted by b_t . The decision to bank permits depends on the following inter-temporal arbitrage condition:

$$b_t \ge 0, \qquad -p_t + \beta \mathbb{E}_t p_{t+1} \le 0 \qquad \text{and} \qquad (-p_t + \beta \mathbb{E}_t p_{t+1}) b_t = 0, \tag{4}$$

⁸We do not need to make assumptions with respect to the properties of B_{θ} and $B_{\theta\theta}$.

⁹Our model does not allow for borrowing of permits from future periods. This constraint is more or less in line with the current practices in the EU-ETS. In addition, throughout the paper, we assume the inter-temporal trading ratio (ITR) to be equal to unity like in Fell et al. (2012).

where $\beta = (1 + r)^{-1}$ is the discount factor and r the periodic interest rate. Banking is not profitable if the expected present value of the next period's permit price is lower than the current permit price. If the opposite holds, banking becomes profitable and, in a competitive equilibrium, all arbitrage possibilities are fully exploited. Thus, banking demand increases the current permit price until the second inequality in (4) holds as an equation.¹⁰

Banked permits contribute to the next period's emissions cap. The regulator allocates Δ_t new permits to the market, totaling in an emission cap, Q_t

$$Q_t := \Delta_t + b_{t-1}.\tag{5}$$

Banking of permits alters the earlier market equilibrium conditions in (3) via the presence of speculative banking demand and banked permits from previous periods. More importantly, possibility to bank permits enforces a strictly positive permit prices.¹¹ The new market equilibrium condition with banking can be written as

$$Q_t - q_t - b_t = 0, (6)$$

where the cap is given by (5). Since borrowing of permits is forbidden the amount of permits banked can be zero or positive. For example, suppose that $b_t = 0$. This means that the current permit price, $p_t > \mathbb{E}_{\approx} p_{t+1}$, is high enough to discourage banking for future periods. If $b_t > 0$, then the current price matches the expected present value of the future permit price, i.e. $p_t = \beta \mathbb{E}_t p_{t+1}$. Thus, the banking motive in effect sets an endogenous price floor for permit prices as the permit price can be higher than the present value of next period price but not lower. The level of the price floor depends on the discount factor and the inter-temporal persistence of shocks, and moreover, it is endogenously intertwined with the regulator's policy function as we will see later. The effect of speculative banking is therefore to transform the cap and trade policy into a version of hybrid policy with a price floor (e.g. Roberts and Spence, 1976), and the regulator can in turn take an advantage of this feature, which we soon demonstrate. But before that, we characterize the price and emissions outcomes in more detail.

There are two possible market outcomes under a permit policy with or without banking: either the emissions are equal to the periodic cap, Q_t or the emissions are lower than the cap. If the emissions are equal to the cap, $q_t = Q_t$, i.e. the cap is binding, then the permit price is simply determined through the market equilibrium condition (2). If the cap is not binding, the actual level of emissions, $q_t < Q_t$, is determined in the market by a given level of cut-off market price of the emission permit, \tilde{p}_t . In the case of no banking, from equation (3) it is obvious that a non-binding cap leads to a permit price equal to zero. Whereas, in the case of banking, a non-binding cap indicates positive banking. The speculators' decision rule (4) determines that the current permit price is equal to the present value of expected next period price of the permit. Thus, the cut-off price levels are $\tilde{p}_t = 0$ and $\tilde{p}_t = \beta \mathbb{E}_t p_{t+1}$

¹⁰Banking can be understood as a nondepreciating storage. Thus, banking decision follows the normal rules of competitive storage, see e.g. Deaton and Laroque (1992) and Amundsen et al. (2006).

¹¹The price is positive only if the allocation of new permits is moderate enough to generate a positive probability for the cap to be binding at some point of time in the future.

for permit market without and with banking, respectively. When inter-temporal trading of permits is allowed, the speculative demand absorbs the supply of excess permits, whereas when banking is not allowed, there is an excess supply of permits and they hence have no value. The resulting market emissions are given by relation

$$q_t = \min\left\{q(\tilde{p}_t, \theta_t), Q_t\right\} \tag{7}$$

and current permit price

$$p_t = \max\left\{\tilde{p}_t, B_q(Q_t, \theta_t)\right\},\tag{8}$$

where the cut-off price level is $\tilde{p}_t = 0$ and $\tilde{p}_t = \beta \mathbb{E}_t p_{t+1}$ for permit market without and with banking, respectively. The market emission function, $q(p, \theta)$, is defined through equation (2).¹²

The current realization of the shock θ_t ultimately determines whether or not the cap is binding. Since high shock realization leads to high emissions, i.e. $q_{\theta}(p, \theta) > 0$, there is a cut-off level of shock realization that divides shock realizations into binding and non-binding regimes. If the realized shock is higher than an endogenously determined cut-off shock level $\tilde{\theta}_t$, the cap is binding. In the opposite case the cap is not binding.

Definition 1. The cut-off value of shock, $\tilde{\theta}_t$, is the shock realization below which, the cap level Q_t is not binding with current price \tilde{p}_t , i.e.

$$\tilde{\theta}_t := \theta(Q_t, \tilde{p}_t),\tag{9}$$

where function $\theta(q, p)$ is implicitly defined by the relation $B_q(q, \theta(q, p)) - p = 0$.

The function $\theta = \theta(q, p)$ states the unique value of shock realization associated with quantity-pricepair (q, p). By the implicit function theorem and given the assumptions on the properties of B, we have: $\theta_q(q, p) > 0$ and $\theta_p(q, p) > 0$. With low shock realizations, $\theta_t < \tilde{\theta}_t$, the cap is not binding and emission and price outcomes are determined through left-hand terms of max and min functions on equations (7) and (8), respectively. With high shock realizations, $\theta_t > \tilde{\theta}_t$, the cap is binding and right-hand terms are active.

The market outcome is determined through equations (7) and (8). When banking of permits is allowed, the cut-off price, $\tilde{p}_t = \beta \mathbb{E}_t p_{t+1}$, is based on the expectations of future market conditions and on the regulator's policy. Thus, the resulting market equilibrium cannot be specified without first examining the regulator's problem.

3 The policy

3.1 Planner's Problem

Emissions from the production of the market good, q, cause environmental damages denoted by an increasing and convex damage function D(q), i.e. $D_q(q) > 0$ and $D_{qq}(q) \ge 0$. A benevolent social

¹²Note that $Q_t \equiv q(B_q(Q_t, \theta_t), \theta_t)$.

planner observes this externality and chooses the level of the policy instrument to maximize the expected social benefits. We next characterize the optimal cap and trade policy in a second-best setting and compare it with the first-best and second-best tax policies. The policy is second-best, since the planner has to set the environmental policy prior to observing the current shock realization (Weitzman, 1974; Yates and Cronshaw, 2001; Feng and Zhao, 2006).¹³ In contrast, the market agents decide their actions after the shock is realized. We assume a Markovian shock process, i.e. $\mathbb{E}_t \theta_{t+1} = \mathbb{E}[\theta_{t+1}|\theta_t].^{14}$

Let us first examine the case of a cap and trade policy with banking. The problem of the regulator is to decide an optimal sequence of new emission permits to be allocated for each period $\{\Delta_t\}_{t=0}^{\infty}$. Given the amount of permits banked in previous period b_{t-1} , the resulting emissions cap for the current period is $Q_t = \Delta_t + b_{t-1}$. The information structure is assumed to be such that when deciding the allocation Δ_t the planner knows the previous period values of all the variables. Especially, the regulator knows the the amount of permits banked, b_{t-1} . This is reasonable as the regulator controls both the allocation of new permits and the invalidation of used permits. We define the information set of the regulator as $\Omega_{t-1} := \{b_{t-1}, \theta_{t-1}\}$. Under this information set, the regulator knows the cap, Q_t , which results from allocating Δ_t new permits. Instead of using the allocation of new permits we formulate the regulator's problem as one of setting the sequence of optimal emission cap levels, $\{Q_t\}_{t=0}^{\infty}$. The optimization problem of planner is dynamic and we formulate it as a dynamic program. We consider only a set of stationary and time-consistent policies. The Bellman equation of the problem is

$$V(b_{t-1}, \theta_{t-1}) = \max_{Q_t} \mathbb{E}_{t-1}[B(q_t, \theta_t) - D(q_t) + \beta V(b_t, \theta_t)],$$
(10)

where emissions q_t are given by the market equilibrium determined in equations (7) and (8), with cut-off price $\tilde{p}_t = \beta \mathbb{E}_t p_{t+1}$. We cannot proceed without first examining the equilibrium emissions.

Analogously to the Deaton and Laroque (1992), the equilibrium of the emission permit markets is a price function $p_t = f(Q_t, \theta_t, b_{t-1})$ implicitly defined by the market price equation (8)

$$f(Q_t, \theta_t, b_{t-1}) = \max \left\{ \beta \mathbb{E}_t f(Q_{t+1}, \theta_{t+1}, b_t), B_q(Q_t, \theta_t) \right\}.$$
 (11)

From here we can observe that the assumed information structure and, therefore, regulator's ability to control the cap levels, makes the amount of permits in the bank, b_{t-1} , redundant. This is a crucial observation on which our approach and results rely on. We next define the reduced equilibrium.

Definition 2. A stationary rational expectations equilibrium is a function $\hat{f}(Q_t, \theta_t)$ defined through

$$\hat{f}(Q_t, \theta_t) = \max\left\{\beta \mathbb{E}_t \hat{f}(Q_{t+1}, \theta_{t+1}), B_q(Q_t, \theta_t)\right\}.$$
(12)

The equilibrium emissions and their properties are given in a following lemma.

Lemma 1. Given an equilibrium $\hat{f}(Q_t, \theta_t)$ the corresponding equilibrium emissions are

$$\hat{q}(Q_t, \theta_t) := \min\left\{q(\beta \mathbb{E}_t \hat{f}(Q_{t+1}, \theta_{t+1}), \theta_t), Q_t\right\}.$$
(13)

 $^{^{13}\}mathrm{Assuming}$ there are no possibilities to design a menu of contingency dependent policies.

¹⁴We use short-hand notation, $\mathbb{E}_t \theta_{t+1} := \mathbb{E}[\theta_{t+1} | \Omega_t]$, where the set Ω_t contains all the information available to the planner in period t. We use the notations for the conditional expectations interchangeably.

The following property holds:

$$\hat{q}_Q(Q_t, \theta_t) = \begin{cases} 0, & \text{if } \theta_t \le \tilde{\theta}_t, \\ 1, & \text{if } \theta_t > \tilde{\theta}_t. \end{cases}$$
(14)

The cut-off shock level is $\tilde{\theta}_t = \theta(Q_t, \beta \mathbb{E}_t \hat{f}(Q_{t+1}, \theta_{t+1}))$ and $\theta_t \leq \tilde{\theta}_t$ and $\theta_t > \tilde{\theta}_t$ denote cases where the cap is non-binding and binding, respectively.

Proof. Appendix A.1.

Although, the market equilibrium is not yet completely solved as the definition (13) is only implicit and the effect of the emission cap policy unspecified, we can proceed in analysis of the regulator's problem. The Bellman equation (10) can be recast as

$$\hat{V}(\theta_{t-1}) = \max_{Q_t} \mathbb{E}_{t-1}[B(\hat{q}(Q_t, \theta_t), \theta_t) - D(\hat{q}(Q_t, \theta_t)) + \beta \hat{V}(\theta_t)],$$
(15)

where market reaction function is given by the equilibrium emissions function (13). Since the only state variable is an exogenous shock, we can restate the problem as a sequence of static optimization problems

$$\max_{Q_t} \mathbb{E}_{t-1}[B(\hat{q}(Q_t, \theta_t), \theta_t) - D(\hat{q}(Q_t, \theta_t))].$$
(16)

Since the emission response function (13) is determined through the market equilibrium (12) the optimization is in effect constrained by the equilibrium function. The optimization leads to a stationary policy that sets the emissions cap for each period, i.e. $Q_t = Q^{bank}(\theta_{t-1})$. In a rational expectations equilibrium, the market agents anticipate correctly the policy rule that sets the cap. Given the information structure of the market agents, at period t they observe the shock realization θ_t and, therefore, know the next period emissions cap, $Q_{t+1} = Q^{bank}(\theta_t)$, too. However, they do not know the next period price of the emission permit as it depends also on the shock realization of that period. In this sense, the problem of the speculative agents is easier under this kind of policy than in the competitive storage setting by Deaton and Laroque (1992). However, the market agents have here less information than in a two-period models of emission permit policy by Yates and Cronshaw (2001) and Feng and Zhao (2006).

It should be noted that if the shock θ_t is identically and independently distributed (IID), the previous shock realization does not convey any information of future shock realizations. Thus, in the IID case, the stationary policy degenerates into static one with single value, i.e. $Q^{bank}(\theta_{t-1}) = Q$, for all θ_{t-1} . In this case the cap is always set to a given level and it is optimal in all the contingencies.

3.2 Characterization of Policy

As stated above, the planner's problem (16) leads to a stationary policy $Q_t = Q^{bank}(\theta_{t-1})$ for the optimal emission cap level. Thus, it is obvious that in a given setting, the optimal cap is independent of the existing bank of permits. We state the implication of this observation in the following proposition:

Proposition 1. The optimal number of new permits allocated completely neutralizes the effect of the existing bank of permits, *i.e.*

$$\Delta_t = \Delta^{bank}(b_{t-1}, \theta_{t-1}) := Q^{bank}(\theta_{t-1}) - b_{t-1}.$$
(17)

Since the planner cares only for the emissions and since the cap, Q_t , is the only instrument in use, it is irrelevant for the planner, whether the cap stems from the newly allocated permits or from the bank. As a consequence, allocation of new permits adjusts fully to neutralize the effect of the existing bank. It is worth noting that the existing literature does not explicitly state the result in Proposition 1. Newell et al. (2005) propose a similar type of a rule, but in that paper the regulator's goal is to fix the permit price to some preferred level in a setting where both banking and borrowing are allowed. Proposition 1 also implies that speculators' demand to bank permits is higher when compared to a situation where the number of permits in the bank increases the level of future permit supply (that is, when Proposition 1 does not hold). To see this, notice that even a large accumulation of banked permits, b_{t-1} , does not decrease the expected future price exactly because of the regulator's active policy.¹⁵ For example, suppose an extreme case where $b_{t-1} > Q^{bank}(\theta_{t-1})$. The planner then adds a negative number of new permits, which in practice entails a purchase of the excess permits off the market.¹⁶ However, as we will later discuss, if an allocation policy of Proposition 1 is implemented, the bank does not become so large that a buy-back would be needed.

In order to examine the policy further, we need to analyze the way the policy is optimally set. We present the necessary first-order condition of the planner's problem (16) in the following proposition:

Proposition 2. The planner's optimal second-best emissions cap policy with banking, $Q^{bank}(\theta_{t-1})$, satisfies

$$\mathbb{E}_{t-1}[B_q(Q_t, \theta_t)|\theta_t > \theta(Q_t, \beta \mathbb{E}_t \hat{f}(Q_{t+1}, \theta_{t+1}))] = D_q(Q_t), \tag{18}$$

for all θ_{t-1} .

Proof. Appendix A.2.

The proposition states that the second-best cap equates the marginal damages to the expected marginal benefits only from those realizations where the emissions are equal to the cap, i.e. when the cap is binding. This follows from the fact that the planner can, at margin, only affect those emission that are bound by the cap. When the cap is not binding, it is the speculative market demand that determines the actual level of emissions. In that case, the market price of emissions is determined by the expectations of future prices, i.e. $p_t = \beta \mathbb{E} p_{t+1}$. These expected prices are, in turn, determined by expectations on price realizations when the cap is binding next time. These expected future prices are, in turn, determined by the same logic by those future price realizations when the cap is binding.

¹⁵The fact that expected future permit price is independent of the current bank of permits simplifies the permit market problem compared to the case of competitive storage models (e.g. Deaton and Laroque, 1992).

¹⁶This is something that the new EU ETS reserve mechanism is planned to execute.

To illustrate the logic of the cap and trade policy with banking, we compare it with second-best cases of cap and trade without banking and tax. A cap and trade policy is naturally very similar to the one with banking. The lack of inter-temporal trading of permits, however, simplifies the problem considerably as we do not need to worry about the equilibrium price function (12). As explained in Section 2, without banking the cut-off price level is exogenously given $\tilde{p}_t = 0$. Without banking it is obvious that the planner can set the cap sequentially as there is nothing that would couple the periods to each other. Thus, we can replace the emission reaction function (13) in the planner's problem (16) with relation $q_t = \min\{q(0, \theta_t), Q_t\}$. Given this market reaction we can state the optimality condition as

$$\mathbb{E}_{t-1}[B_q(Q_t, \theta_t)|\theta_t > \theta(Q_t, 0)] = D_q(Q_t).$$
(19)

As in the case of cap and trade with banking, the second-best cap equates the marginal damages to the expected marginal benefits only from those realizations where the cap is binding. Thus, it seems to be a general property of the cap and trade policy. In the case of no banking, if the shock is limited to be such that the cap is always binding, we no longer need the zero price floor in the first-order condition (cf. Feng and Zhao, 2006, eq. (3)). Our result extends the optimal policy to the case where economic shocks can be so large that emissions cap is not always binding. While the possibility of non-binding cap seems to be a rather rare special case, the non-binding limit becomes a fundamental property of the policy when banking of permits is allowed. Proposition 2 characterizes the planner's optimal management of permit markets with banking and equation (19) without banking, and to our knowledge, they have not been stated in the literature before.

The problem of setting the second best tax can again be performed sequentially. The planner sets the optimal tax based on the following optimization problem:

$$\max_{\tau_t} \mathbb{E}_{t-1}[B(q(\tau_t, \theta_t), \theta_t) - D(q(\tau_t, \theta_t))].$$
(20)

Unlike in the case of cap and trade, here the emission outcome is directly obtained from (2) with tax being the price of emission. The necessary first-order condition to this problem is

$$\mathbb{E}_{t-1}[B_q(q(\tau_t,\theta_t),\theta_t)q_p(\tau_t,\theta_t)] - \mathbb{E}_{t-1}[D_q(q(\tau_t,\theta_t))q_p(\tau_t,\theta_t)] = 0.$$
(21)

Since $B_q(q_t, \theta_t) = \tau_t$ by equation (2), we get

$$\tau_t \mathbb{E}_{t-1}[q_p(\tau_t, \theta_t)] - \mathbb{E}_{t-1}[D_q(q(\tau_t, \theta_t))q_p(\tau_t, \theta_t)] = 0$$
(22)

The above equation implicitly determines the second-best optimal tax level, $\tau_t^{2nd}(\theta_{t-1})$. Term $q_p(\tau_t, \theta_t) = B_{qq}^{-1}(q_t, \theta_t)$ describes the changing curvature of the marginal benefit function. The curvature term in the first-order condition gives more weight to those contingencies where the constant tax forces a larger change in the production level.¹⁷ Comparison of (22) with optimality conditions of cap and trade policies (18) and (19) non-linearities in the model have a more direct effect on optimal policy

¹⁷The past literature has typically analyzed the case of quadratic benefit and damage functions. In that case the curvature term is constant and has no effect on the solution of the problem.

in the case of tax than with cap and trade policies. In the case of tax, the lack of marketable permits enables simple solutions in special cases: If the marginal damages are constant, $D_q(q) = d$ for all q, then $\tau_t = d$, for each period as the curvature terms cancel out. In addition, if the benefit function is quadratic, the curvature term is constant and tax is implicitly determined by $\tau_t = \mathbb{E}_{t-1}D_q(q(\tau_t, \theta_t))$.

Allowing the inter-temporal trading introduces a new parameter into the model: the discount factor β . The discount factor has an important role in the determining the banking motives (4) and, therefore, the endogenous price floor. The smaller the discount factor, the weaker the banking motives, as expected relative price increases need to be higher to induce banking. In fact, if the discount factor is low enough, there may be situations where banking has no role as $p_t > \beta \mathbb{E}_t p_{t+1}$ always holds. More precisely, we define the critical level of the discount factor to be such that there will be no demand for speculative banking.

Definition 3. The no-banking critical level of discount factor is

$$\beta_0 := \sup\{\beta \mid p(Q_t^N, \theta_t) > \beta \mathbb{E}_t p(Q_{t+1}^N, \theta_{t+1}), \forall \theta_t \in S_{\theta_t}\}$$

where Q_t^N is the optimal emission cap without banking and S_{θ_t} denotes the support of probability distribution of θ_t .

The above definition of the critical discount factor directly leads to the following proposition.

Proposition 3. Zero amount of permits is banked, $b_t = 0$, and banking has no role in a cap and trade policy if the discount factor is less than the critical level, i.e. $\beta < \beta_0$.

Proof. The definition of critical discount factor, β_0 , presents the largest value of discount factor that cannot support positive banking.

Whether the no-banking result is probable in applications depends on the stochastic process of the shock θ_t and the length of the period in a discrete time setting. For example, there is no banking if the periodic interest rate is higher the largest expected relative price increase.¹⁸ If the underlying shock process has a wide enough support, banking always has at least a potential effect on the optimal cap. In the opposite extreme of a deterministic model, the policy can be set optimally for each period and there is no need for banking demand.

The value of the discount factor, β , relative to the critical discount factor, β_0 , determines how the price floor influences the planner's optimal policy. The following proposition shows how the optimal level of the cap is set in the presence of banking demand:

Proposition 4. Let $B_{q\theta}(q,\theta) > 0$, $B_{qq}(q,\theta) < 0$ and $D_{qq}(q) \ge 0$, for all q and θ . Given an optimal cap without and with banking Q_t^N and Q_t^B , respectively. If $\beta < \beta_0$, then $Q_t^B = Q_t^N$ for all t. However, $\frac{if \ \beta > \beta_0, \ then \ Q_t^B > Q_t^N, \ for \ all \ t.}{^{18}\beta < \beta_0 \Leftrightarrow r > \mathbb{E}_t \max\{(p_{t+1} - p_t)/p_t\}.}$



Figure 1: A schematic illustration of the effect of banking on the emission permit markets with IID shocks. Gray lines $\mathbb{E}B_q$ and D_q are the expected marginal benefits and marginal damages, respectively. τ denotes the optimal tax and Q^N denotes the optimal cap under no banking. Optimal cap with banking is denoted by Q^B and endogenous price floor is given by $\beta \mathbb{E}p_{t+1}$. The market outcome is at intersection of the actual realization of B_q and tax-level (τ), cap-level (Q^N) or policy locus (red dashed line) in the case of tax, cap without and with banking, respectively.

Proof. Appendix A.3

Proposition 4 states that if the discount factor is less than the critical level, there will be no motive to bank permits for future periods, and therefore, option to bank permits will not influence the planner's optimal policy. If, on the other hand, the discount factor is large enough, then it is possible that positive banking occurs under some shock realizations. In that case, the planner utilizes the presence of an endogenous price floor by increasing the level of the optimal cap relative to no-banking case $(Q^B > Q^N)$. The reason for this is that the presence of a price floor reduces the efficiency losses incurred at low shock levels. This in turn enables the planner to hedge against efficiency losses at high shock levels by relaxing the cap. Thus, the active management of the permit allocation together with the planner's policy to neutralize the effect of the bank result in differing cap levels under banking and no banking. This is in contrast to the optimal allocation rule presented by Feng and Zhao (2006) where the cap levels are equal under banking and no banking. This difference is due to the regulator's ability to actively adjust the cap level based on the observed market outcome.

Figure 1 illustrates the effect of banking on optimal cap level. The endogenous price floor generated by the speculative demand, in effect, creates a policy instrument that combines the properties of the quantity and price policies. When the realized shock is high and the cap is binding, we have a quantity regulation in place, whereas when the realized shock is low and the cap is not binding, we have a price policy in place. The price floor can be above or below the optimal 2nd best tax level depending on the model parameters. In the case of no banking, the determination of the optimal cap level is rather straightforward for the planner as the zero price floor is exogenously given. When banking is allowed, however, the problem becomes more difficult. In that case, the planner has to satisfy the first order condition (18) and also assess the market equilibrium price function (12) for the emissions permits. The difficulty resides in the two-way connection between the price function and the optimal policy: market agents adapt to the policy and the policy needs to adapt to the market behavior. While the banking combines features of the price and quantity regulation (Figure 1), the price floor itself emerges through the speculators demand in the permit market. Thus, the planner has only a partial control over how the price-quantity combination is fulfilled.

4 Results

4.1 IID shocks

We first analyze the planner's optimal policy in a special case where the periodic random shocks, θ_t , are independent and identically distributed. The next subsection examines the effect of shock persistence. We can use our model to derive an analytical solution under a set of assumptions. Namely, the benefit and damage functions are quadratic polynomials, the optimal cap is always binding without banking and the IID shock follows a continuous, uniform distribution $\theta \sim U[\underline{\theta}, \overline{\theta}]$. The quadratic form of the equations is a standard in a prices vs. quantities literature initiated by Weitzman (1974). The analysis here extends the literature into the case of bankable quantities. The assumption on a binding cap is reasonably minor constraint for feasible parameter values and is made to keep notation as easy to follow as possible. In what follows, we denote the level of dispersion with σ , defined as one half of the width of the support, i.e. $2\sigma = \overline{\theta} - \underline{\theta}$. The uniform distribution allows for simple presentations for expected values and are needed for solving the model analytically. Formally, we assume linear marginal damage

$$D_q(q) = d_0 + D_{qq}q \tag{23}$$

and marginal benefit functions

$$B_q(q,\theta) = \theta - |B_{qq}|q, \tag{24}$$

where parameters $D_{qq} \ge 0$ and $B_{qq} < 0$ denote the slopes of the marginal damage and benefit lines, respectively.

Under the above assumptions, the analytical solution for the optimal policy, with or without banking, is determined by a root of a quadratic polynomial. See Appendix B for details. It is now relatively straightforward to perform comparative statics on the optimal emission cap. We collect the main results in the following proposition.

Proposition 5. Given that there is a positive probability for banking, the optimal cap Q^B increases both with dispersion σ and discount factor β .



Figure 2: The optimal policy for three values of relative steepness of marginal damages $|D_{qq}/B_{qq}| \in \{0, 1, 10\}$. Policy locus is described for three levels of discount factor, $\beta \in \{0.925, 0.975, 0.999\}$, red solid, broken and dotted line, respectively. Black solid line is the mean marginal benefit function and dotted ones are the minimum and maximum realizations of a uniform distribution. Black broken line is marginal damage function. The optimal no banking cap $Q^N = 0.5$ and optimal second best tax $\tau = 0.5$ in all the cases.

Proof. Direct application of comparative statics. See Appendix B.3 for details.

As discussed earlier, since demand for banking generates an endogenous price floor, the planner can reduce efficiency losses at higher shock realizations by relaxing the cap. The planner hence raises the cap if there is an increase in volatility. Similarly, since the demand for banking is increasing in the discount factor, thus making the probability of banking larger, the optimal policy focuses on reducing efficiency losses at high emission realizations by relaxing the cap.

Figure 2 illustrates the determination of optimal policy when banking of permits is allowed.¹⁹ The figure contains three different cases for the ratio of slopes of the marginal damages to marginal benefits, $|D_{qq}/B_{qq}| \in \{0, 1, 10\}$, and for each such case, we illustrate the effect of the discount factor under three different discount factors $\beta \in \{0.925, 0.975, 0.999\}$.²⁰ The vertical red lines represent the levels of the optimal caps under different discount factors whereas the kink point represents the endogenously

¹⁹In the calculations we have assumed that $\mathbb{E}\theta = 1$ and $|B_{qq}| = 1$. Thus, in the market equilibrium without policies, the expected emission level would be unity. The slope of marginal damages, D_{qq} , is varied and the level, d_0 , is chosen to be such that with all the slope values the no banking optimal cap is $Q^N = 0.5$. The dispersion parameter is $\sigma = 0.05$ and resulting $\beta_0 = 0.9$.

²⁰The ratio of slopes is a known determinant of relative performance of price and quantity policies: quantity regulation is better when ratio is large and price regulation when the ratio is small. In the case of bankable permits, however, the endogenous banking motive has an effect on the performance of the quantity regulation. The strength of the banking motives is illustrated here through the discount factor β . Under the assumed parameter values, the discount factor $\beta = 0.9$ is equal to the case of cap and trade without banking. As the discount factor approaches unity, the probability of positive banking increases to the limiting value of unity. The optimal cap in the no-banking case is $Q^N = 0.5$ and the optimal second best tax is $\tau = 0.5$ in all cases.



Figure 3: Optimal cap Q, expected banking relative to the cap $\mathbb{E}b/Q$, expected emissions $\mathbb{E}q$ and expected permit price $\mathbb{E}p$ as a function of discount factor. Black solid, broken and dotted line present $|D_{qq}/B_{qq}|$ values of 0, 1 and 10, respectively. No banking cap $Q^N = 0.5$ and optimal second best tax $\tau = 0.5$ in all the cases. Here $\sigma/D_q(Q^N) = 0.1$.

determined price floor where banking demand becomes positive. The horizontal and vertical parts of the red lines form the policy locus. The intersection of the ex-post marginal benefit line with the policy locus determines the actual level of emissions and the current permit price. Thus, in order to assess the possible price-emissions outcomes under cap and trade policy, the policy locus is substituted for the marginal damage curve.

Figure 2 shows how the presence of an endogenous price floor imposes a restriction on the planner. In the first-best optimum, the policy would be set at the crossing of the ex-post marginal benefit and damage curves. Under the second-best policy, the equilibrium level of emissions is determined through the intersections of the policy locus (red lines) and ex-post marginal benefit curve. The policy therefore performs better the closer it can follow the marginal damage curve. In the case where $|D_{qq}/B_{qq}| = 0$, banking allows the policy to approach the marginal damage curve, especially if the discount factor is close to unity. With steep marginal damages ($|D_{qq}/B_{qq}| = 10$), banking seems to distort the policy away from the marginal damages. In an intermittent case the kinked policy is never an excellent approximation of the marginal damages but may yield welfare gains compared to the quantity without banking and price regulations.

Figure 3 presents the optimal emission cap and the resulting market equilibrium as functions of



Figure 4: Social welfare gain relative to the cap and trade without banking as a function of discount factor β . Solid black line is the optimal cap and trade with banking and dashed red line is the optimal tax. With the lowest value of $\beta = 0.9$ there is no banking in the equilibrium and, thus, banking and no banking cases are equal. The red circle shows the maximal welfare under cap and trade with banking. Here $\sigma/D_q(Q^N) = 0.1$.

the discount factor β . We again compare three different slope ratios $(|D_{qq}/B_{qq}| \in \{0, 1, 10\})$. While the emission cap increases with the discount factor the expected banking increases too, resulting in non-monotonic expected emissions. The resulting expected permit price pattern mirrors that of the expected emissions. As Figure 2 already implied, the planner allows the cap to vary more when the marginal damages are flat. When the marginal damages are steep, the optimal cap is hardly changing at all. When compared to a permit system without banking, with steep marginal damages, banking tends to decrease expected emissions, whereas with flat damages, banking increases expected emissions. With intermediate slopes, expected emissions increase and decrease with low and high values of discount factor, respectively. The expected share of emission to be banked does not depend on the relative slopes of the marginal damages and benefits. Instead, the banking increases with discount rate in all the three cases studied.

Figure 4 compares the social welfare under three different instruments: permit system without banking (reference), permit system with banking (black solid line) and the optimal second-best tax (red dashed line). The comparison is measured as a gain in expected social welfare relative to the expected social welfare from using the cap and trade without banking. The figure shows that relative gains or losses in welfare depend on the discount factor β . Left panel shows that allowing for banking of permits increases welfare when the marginal damages are flat. In the extreme limit of $\beta \rightarrow 1$ the welfare gain approaches the gain from using a tax, which is known to be efficient policy is this case. This results from the fact that in the limiting case, banking is effectively equal to the tax policy. For other case of $|D_{qq}/B_{qq}|$ optimal banking policy increases welfare, although as the ratio increases the gains decrease. When the marginal damages are very steep and the discount factor high, banking can worsen the performance of permit regulation, resulting in relatively high welfare losses. These results are different to the result found in Feng and Zhao (2006, Proposition 2). They observe that with ITR equal to unity, banking

will decrease welfare if $D_{qq} > |B_{qq}|$. We show here that if the allocation of new permits is actively managed, banking can increase expected welfare even when $D_{qq} > |B_{qq}|$. If the difference in slopes is large, the magnitude of gains becomes smaller and less likely (panel $|D_{qq}/B_{qq}| = 10$). However, if the difference of slopes is smaller, i.e., situation between panels $|D_{qq}/B_{qq}| = 1$ and $|D_{qq}/B_{qq}| = 10$, then the welfare gains are larger. All in all, the welfare differences are rather minor between any second-best regulation. The relative differences seem to be limited to 1 percent or less. Yet, the absolute differences may be notable, especially, with global externalities such as greenhouse gas emissions.

It is interesting to note that the discount factor is not necessarily exogenous from the perspective of the planner. By shortening the interval of the permit adjustments the discount factor can be increased. In principle, by a proper choice of period length any discount factor can be reached. Thus, the model suggests that with a very flat marginal damages, the period should be as short as possible for the permit system to approach the optimal performance of the tax regulation. With very steep marginal damages, the bankable permits do not seem to be useful unless a very long period can be enforced. It is worth noting, that the period length is typically positively correlated with dispersion of the cost shock. The effect of dispersion is not taken into account in the above discussion.

4.2 Persistent shocks

To study the optimal management of permit allocation over business cycles, we need to examine the case of persistent shocks. When the shock exhibits persistence, the expected future periods are not identical conditional on the most current information. Therefore, both the optimal policy and the optimal banking decisions by the speculative agents vary between periods. Solving the optimal policy becomes notably more difficult when compared to the case of IID shocks. Yet, highly correlated economic shocks can create persistent anomalies in the demand for permits, and failure to manage the allocation of new permits may lead to much greater deviation from the optimal emissions level compared to the case of IID shocks. Thus, with persistent shocks there is a greater need for optimal allocation rules for new permits.

We are interested in how the optimal emission cap is set under persistent shocks when banking of permits is allowed. It turns out that the effects of the discount factor and of the slopes of the marginal damage and benefit functions do not differ that much from the IID case above. In the end, the planner and the speculative agents face a similar Markov decision problem in both cases. Only significant difference is that, with persistent shocks, the current realization yields useful information with respect to the future realizations. Apart from this, there is little difference in periodic decision making.

We solve the optimization model numerically through a iterative scheme (Appendix C). The Markov process for the shock is specified as an autoregressive AR(1) process

$$\theta_t = (1 - \varphi) \mathbb{E}\theta + \varphi \theta_{t-1} + \varepsilon_t,$$



Figure 5: The optimal emission cap for four values of persistence $\varphi \in \{0, 0.4, 0.7, 0.9\}$ with dotted, broken and dotted, broken and solid lines. The relative steepness of marginal damages $|D_{qq}/B_{qq}| = 1$ and the discount factor, $\beta = 0.95$.

where innovations are IID Gaussian with $\varepsilon_t \sim N(0, \sigma_{\varepsilon}^2)$. In what follows, we assume that $\mathbb{E}\theta = 1$. For the optimization problem, we discretize the stochastic process into about a hundred states and describe the Markov process through the transition matrix P_{ij} . Here we concentrate on persistence levels $\varphi \in \{0, 0.4, 0.7, 0.9\}$, where zero case represents the IID case.²¹ In addition, both the B_{qq} and D_{qq} are set to unity and the discount factor $\beta = 0.95$. We have set the unconditional standard deviation of the θ_t to be $\sigma_{\theta} = 0.05$. Therefore, the conditional standard deviation for the next period prediction is $\sigma_{\varepsilon} = (1 - \varphi^2)\sigma_{\theta}$ indicating a lower risk level for the regulator as the persistence φ becomes stronger.

Figure 5 presents the relationship between the optimal emission cap for the next period and the observed current realization of the shock term, θ_{t-1} , i.e. $Q_t = Q^*(\theta_{t-1})$. We have used four different values of persistence $\varphi \in \{0, 0.4, 0.7, 0.9\}$. As can be seen, the emissions cap is increasing in the value of shock variable. The higher the persistence the more strongly the cap responds to the current economic conditions. As the persistence becomes weaker the optimal policy approaches the constant IID case. With high shock realizations, the optimal cap locus is reasonably linear, but as the shock value decreases, an increasing convexity is observed.

Figure 6 illustrates the effect of optimal cap on market outcomes. The red curves show the quantityprice pairs that result in a permit system with banking. The figure presents three different shock realizations $\theta_{t-1} \in \{0.85, 1, 1.15\}$, where unity is the process mean and 0.85 and 1.15 are very low and very high realizations, respectively. Depending on the level of persistence, the expected value of the next period's shock is somewhere between the process mean and the observed value. Comparing the Figure 6 with the IID case in Figure 2, we can see that the horizontal portion of the policy, where banking demand is positive, is actually becoming lower as the emissions decrease. This indicates that

 $^{^{21}}$ Here we have a normal distribution for the shock and, thus, the results are not identical with the IID case in the previous section.



Figure 6: The optimal cap and trade policy with banking for three values of persistence $\varphi = 0.4$ (left), $\varphi = 0.7$ (middle) and $\varphi = 0.9$ (right panel). Policy loci for three shock realizations on the previous period, $\theta_{t-1} \in \{0.85, 1, 1.15\}$, are presented by red solid, broken and dotted lines, respectively. The black lines describe the marginal benefit curve with the three cases of θ_{t-1} . Blue solid line is the marginal damage function. The relative steepness of marginal damages $|D_{qq}/B_{qq}| = 1$ and the discount factor $\beta = 0.95$.



Figure 7: The expected emissions (left panel), permit price (middle panel) and banking (right panel) for four values of persistence $\varphi \in \{0, 0.4, 0.7, 0.9\}$ with dotted, broken and dotted, broken and solid lines. The relative steepness of marginal damages $|D_{qq}/B_{qq}| = 1$ and the discount factor $\beta = 0.95$.

the expected permit price is lower for low shock realization. The reason for this is that the demand for banking also decreases as the speculators expect future permit prices to remain depressed. This price dependence is stronger with higher levels of persistence. Thus, when persistence becomes weaker, we gradually approach the IID case in Figure 2.

Figure 7 shows the expected values of emissions, permit price, and banking as functions of the realized shock level. With extreme shock realizations, expected emissions are highest when persistence is also high. With shock realizations near the mean, low persistence cases have the highest expected emissions. Expected price level mirrors those of emissions. Expected banking is positive only with low shock realizations, and banking demand is highest when persistence is weak. With high persistence a low shock realization is most likely reached from a previous low shock realization, which indicates that



Figure 8: The optimal policy locus of cap and trade with banking for three values of shock realizations on the previous period, $\theta_{t-1} \in \{0.85, 1, 1.15\}$, are presented by red solid, broken and dotted lines, respectively. The black lines describe the marginal benefit curve with three cases of θ_{t-1} . Blue solid line is the marginal damage function. The persistence of shock is $\varphi = 0.9$. The relative steepness of marginal damages $|D_{qq}/B_{qq}| = 0$ and the discount factor, $\beta = 0.95$.

the cap level is already low, as the regulator anticipates the persistence of low shocks (see Figure 5). If the persistence is low, the cap level is relatively higher as the cap is varied less. In addition, with low persistence the low shock realization is more likely reached from a higher realization, under which a higher cap is set. Both of these effects make the likelihood of excess supply of permits higher and this induces strong banking motives.

4.3 Case: climate change

Climate change is driven by the atmospheric carbon stock. Since the emitted CO₂ particles have long life spans in the atmosphere, the current atmospheric carbon stock is an outcome of centuries of inflow. This leads to a very flat marginal damages curve (Hoel and Karp, 2002; Newell and Pizer, 2003). This observation is based on the fact that the marginal damages from carbon emissions derive from a large atmospheric carbon stock which is in turn driven by the flow of annual emissions. Thus, the effect of a small variation in annual emissions has a negligible effect on the carbon stock and therefore to marginal damages.²² Although we do not have stock variables in our model, we can examine CO₂ cap and trade policies by assuming that the slope of the marginal damages D_{qq} is zero, i.e., the period-wise damages are linear.

Building on this observation, we assume that the periodic marginal damages from the emissions are

 $^{^{22}}$ Note that the constant periodic marginal damages do not need to be time invariant.



Figure 9: The optimal cap (a) and expected banking (b), emissions (c) and permit price (d). The persistence of shock is $\varphi = 0.9$. The relative steepness of marginal damages $|D_{qq}/B_{qq}| = 0$ and the discount factor, $\beta = 0.95$.

constant. As stated in Section 3.1 the optimal second best tax is equal to the marginal damages $\tau = MD$. The welfare analysis in Section 4.1 suggests that the tax would be an optimal second best policy and that banking will strongly improve the welfare impact of a cap and trade policy. To simulate the business cycles and the carbon emission policy, we assume that the shock is highly persistent ($\varphi = 0.9$). The analysis proceeds along the same lines as in the previous section. Figure 8 shows the policy loci under an optimal cap and trade policy with banking. The loci are presented for the cases of low, mean and high observed shock realizations θ_{t-1} . The form of the policy locus is similar to the IID case as the permit price is not affected by the shock realization when there is positive banking, i.e. the flat portion of the policy locus is horizontal.

Figure 9 presents the optimal cap for the next period given the observed shock realization θ_{t-1} (panel a). Due to the quadratic benefits and constant marginal damages, the relationship is linear. Because of this linearity, under the optimal policy, the expected permit price, $\mathbb{E}[p_{t+1}|\theta_t]$, is constant irrespective of the shock realization. The constancy of ex-ante expected permit price explains the horizontal policy locus seen in Figure 8. Panels b, c and d of Figure 9 show the expected banking, emissions, and permit prices, respectively, for each shock realization. The banking activity is similar to the case of quadratic damages (Figure 7). Banking is profitable only if shock decreases strongly enough. Therefore, banking is observed only with low shock realizations. There is a notably larger variation in the optimal cap and

resulting emissions (panesl a and c). This occurs because the optimal policy tries to keep the permit price as close to the marginal damages function as possible. Since the marginal damages are flat, the variation in expected prices is quite modest over the business cycle. The expected price realizations, $\mathbb{E}[p_t|\theta_t]$, are positively correlated with the shock, even though, the ex-ante expectations $\mathbb{E}[p_{t+1}|\theta_t]$ are constant. This results from the fact that the high shock realizations are more likely generated by positive innovations that lead to binding cap. The opposite holds for the low realizations where banking demand weakens the decrease in expected prices.

Our results suggest that over the business cycles the cap should be set so that the future expected price would be equal to the assessed social cost of carbon for that period. This was also the basis of analysis in Newell et al. (2005). Under the second-best policy described here, this would lead to higher expected emissions but also higher expected permit prices when the economic activity is high. Yet, the price changes over time would be modest. The results from macroeconomic modeling of the first-best carbon tax indicate that both the emissions and the tax levels could, typically, be increased during economic boom (Heutel, 2012; Lintunen and Vilmi, 2013). In the second-best setting above, the macro economic effects would make the connection between the realized shock level and both the expected emissions and expected prices stronger.

5 Conclusions

This paper has examined the question of an optimal management of permit markets in a secondbest setting where the regulator sets the policy before the cost shock is realized and the market agents act after the realization. We have focused on the case where banking of permits is allowed, but not borrowing, and the inter-temporal trading ratio is unity. We have derived the planner's optimality conditions and characterized analytically the effects of banking on the planner's optimal policy. Under optimal policy, the regulator sets the cap based on expectation on future economic activity. In particular, this means that the current bank level does not affect the choice of the next period's cap level. In a sense, the regulator neutralizes the effect of the existing bank by correspondingly adjusting the allocation of new permits. In most relevant cases, the cap is set at a higher level than in permit systems without banking. This differs from earlier results where there is no active management of permit allocation (Yates and Cronshaw, 2001; Feng and Zhao, 2006). We have also shown that the presence of speculative banking demand sets an endogenously determined price floor which critically depends on the discount rate, persistence of the business cycle, and on the planner's policy.

With IID shocks and under standard assumptions, we show that the optimal cap increases with the discount factor and with shock dispersion. As the optimal policy maximizes net benefits, the optimal cap is driven by the ratio of the slopes of the marginal damage and marginal benefit functions as in Yates and Cronshaw (2001) and Feng and Zhao (2006). We also compare the levels of social welfare under permit systems with banking to those without banking and second-best taxes. We show that when marginal damages are relatively flat, an optimally managed permit system with banking approaches the efficiency of a tax instrument as the time between updates becomes shorter. When marginal damages are relatively steep, too frequent updating of the policy actually decreases welfare relative to permit systems without banking. We show that banking of permits can be welfare improving also when the marginal damage function has a greater slope than the marginal benefit function.

Our numerical analysis also accommodates persistence in the shock process. We show that higher persistence translates to greater variance in the optimal cap, as the planner uses information provided by the observed shock to update expectations of future emissions and permit prices. Likewise, speculative banking demand responds to persistence in shocks by adjusting the price floor to accommodate information provided by the observed shocks. Finally, we use our model to examine the optimal management of present-day cap and trade systems such as EU ETS. We show that in the presence of persistent business cycles, the planner acting optimally makes considerable adjustments to the periodic cap while keeping the expected permit price at a constant level. During economic downturns, the cap is tightened, and during upturns, it is relaxed. Adjustments are done in such a manner that during periods of high economic activity, the expected permit price is higher than in periods of low activity.

Our findings suggest that the permit price should remain relatively stable over the business cycles. Even persistent periods of low shocks should not decrease the permit price significantly. Thus, the findings imply that the prevailing situation in EU emission trading system with high number of permits in the bank and correspondingly low level of permit price is not optimal. Our results indicate that the cap and trade policy should be based on an allocation rule such that the existing bank of permits would decrease the allocation of new permits one-to-one and the emission permit price should remain close to an assessed target level.

The literature on permit systems with banking has also examined the effects of market power on efficiency and on the path of emissions (Liski and Montero, 2005, 2006, 2011). Some firms, due to their size or because of thinness of markets, may be able to influence the market price to their own advantage. A regulator managing the market in such a context needs to adjust the allocation rule accordingly. We leave the determination of the exact nature of this adjustment for future research.

Acknowledgements

Funding from the Academy of Finland grant no. 257919 is gratefully acknowledged. The authors also wish to thank Matti Liski for constructive discussions and Pauli Lappi for providing useful comments at the 37^{th} Annual Meeting of the Finnish Economic Association.

References

- Amundsen, E. S., Baldursson, F. M., and Mortensen, J. B. (2006). Price volatility and banking in green certificate markets. *Environmental and Resource Economics*, 35(4):259–287.
- Chevallier, J. (2012). Banking and borrowing in the eu ets: A review of economic modelling, current provisions and prospects for future design. *Journal of Economic Surveys*, 26(1):157–176.
- Cronshaw, M. B. and Kruse, J. B. (1996). Regulated firms in pollution permit markets with banking. Journal of Regulatory Economics, 9(2):179–189.
- Deaton, A. and Laroque, G. (1992). On the behaviour of commodity prices. The Review of Economic Studies, 59(1):1 – 23.
- Fell, H., MacKenzie, I. A., and Pizer, W. A. (2012). Prices versus quantities versus bankable quantities. *Resource and Energy Economics*, 34(4):607–623.
- Feng, H. and Zhao, J. (2006). Alternative intertemporal permit trading regimes with stochastic abatement costs. *Resource and Energy Economics*, 28(1):24–40.
- Heutel, G. (2012). How should environmental policy respond to business cycles? Optimal policy under persistent productivity shocks. *Review of Economic Dynamics*, 15(2):244–264.
- Hoel, M. and Karp, L. (2002). Taxes versus quotas for a stock pollutant. Resource and Energy Economics, 24(4):367–384.
- Judd, K. L. (1998). Numerical methods in economics. MIT press.
- Kling, C. and Rubin, J. (1997). Bankable permits for the control of environmental pollution. *Journal* of *Public Economics*, 64(1):101–115.
- Leard, B. (2013). The welfare effects of allowance banking in emissions trading programs. *Environ*mental and Resource Economics, pages 1–23.
- Lintunen, J. and Vilmi, L. (2013). On optimal emission control Taxes, substitution and business cycles. Bank of Finland Research Discussion Paper 24/2013.
- Liski, M. and Montero, J.-P. (2005). A note on market power in an emission permits market with banking. *Environmental and Resource Economics*, 31(2):159–173.
- Liski, M. and Montero, J.-P. (2006). On pollution permit banking and market power. Journal of Regulatory Economics, 29(3):283–302.
- Liski, M. and Montero, J.-P. (2011). Market power in an exhaustible resource market: The case of storable pollution permits^{*}. *The Economic Journal*, 121(551):116–144.
- Newell, R. and Pizer, W. (2003). Regulating stock externalities under uncertainty. Journal of Environmental Economics and Management, 45(2):416–432.

- Newell, R., Pizer, W., and Zhang, J. (2005). Managing permit markets to stabilize prices. Environmental and Resource Economics, 31(2):133–157.
- Roberts, M. J. and Spence, M. (1976). Effluent charges and licenses under uncertainty. Journal of Public Economics, 5(3):193–208.
- Rubin, J. D. (1996). A model of intertemporal emission trading, banking, and borrowing. Journal of Environmental Economics and Management, 31(3):269–286.
- Schennach, S. M. (2000). The economics of pollution permit banking in the context of title iv of the 1990 clean air act amendments. *Journal of Environmental Economics and Management*, 40(3):189– 210.
- Weitzman, M. L. (1974). Prices vs. quantities. The review of economic studies, 41(4):477-491.
- Williams, R. (2002). Prices vs. quantities vs. tradable quantities. NBER working papers, 9283.
- Yates, A. J. and Cronshaw, M. B. (2001). Pollution permit markets with intertemporal trading and asymmetric information. Journal of Environmental Economics and Management, 42(1):104–118.

A Proofs

A.1 Proof for Lemma 1

Proof. The equilibrium emissions follow directly from equation (7) and equilibrium (12). Suppose that $\theta_t \leq \tilde{\theta}_t$. Then the cap is not binding and has no effect on the equilibrium emissions level q_t . Suppose that $\theta_t > \tilde{\theta}_t$. Then the cap is binding, $q_t = Q_t$, and the cap has a one-to-one effect on the equilibrium emissions.

A.2 Proof for Proposition 2

Proof. By the Lemma 1 we know that the function \hat{q} is differentiable almost everywhere and the zeromeasure point of non-differentiability does not contribute to the expectation value as the function \hat{q} is continuous. Thus, the first-order condition is directly obtained

$$\mathbb{E}_{t-1}[B_q(q_t, \theta_t)\hat{q}_{Qt}] - \mathbb{E}_{t-1}[D_q(q_t)\hat{q}_{Qt}] = 0,$$
(25)

where we have used the following shorthand notation: $q_t := \hat{q}(Q_t, \theta_t)$ and $\hat{q}_{Qt} := \hat{q}_Q(Q_t, \theta_t)$.

Further using the Lemma 1 we know that $\hat{q}_{Qt} = 1$, if $\theta_t > \tilde{\theta}_t := \theta(Q_t, \tilde{p}_t)$ and zero elsewhere. In addition, $\hat{q}_t = Q_t$, if $\theta_t > \tilde{\theta}_t$. Using these facts we can rewrite the optimality condition

$$\int B_q(Q_t, \theta_t) \mathbf{1}_{\text{BIND}}(\theta_t) dF(\theta_t, \theta_{t-1}) = D_q(Q_t) \int \mathbf{1}_{\text{BIND}}(\theta_t) dF(\theta_t, \theta_{t-1}),$$

where $F(\theta_t, \theta_{t-1})$ is the distribution function of the Markovian shock process. Indicator function $\mathbf{1}_{\text{BIND}}(\theta)$, has value one when $\theta \in \text{BIND}$ and zero elsewhere. The set BIND consists of shock realizations in which the emissions are bound by the cap, i.e. $\text{BIND} := \{\theta_t | \theta_t > \tilde{\theta}_t\}$. Thus, we can write

$$\int_{\text{BIND}} B_q(Q_t, \theta_t) dF(\theta_t, \theta_{t-1}) = D_q(Q_t) \int_{\text{BIND}} dF(\theta_t, \theta_{t-1}) dF(\theta_t$$

Since $\int_{\text{BIND}} dF(\theta_t, \theta_{t-1}) = Pr(\theta \in \text{BIND})$, this directly leads to the optimality conditions

$$\mathbb{E}_{t-1}[B_q(Q_t, \theta_t)|\theta_t > \theta(Q_t, 0)] = D_q(Q_t), \tag{26}$$

for the cap and trade without banking and

$$\mathbb{E}_{t-1}[B_q(Q_t, \theta_t)|\theta_t > \theta(Q_t, \beta \mathbb{E}[p_{t+1}|\theta_t])] = D_q(Q_t),$$
(27)

with banking. The future price is determined through equilibrium price function $p_{t+1} = p(Q_{t+1}, \theta_{t+1})$. Note that the condition with banking has θ_t on both sides of the inequality and, therefore, the conditioning is in implicit form.

A.3 Proof for Proposition 4

Proof. Since $\beta > \beta_0$, with fixed cap level Q_t^N there is a smaller set of shock realizations θ_t which cause the cap to be binding with banking than without banking. This is because in some realizations banking motives are realized. As we have assumed $B_{q\theta} > 0$, this means that the cut-off shock realization with banking is higher than without, i.e. $\tilde{\theta}^B > \tilde{\theta}^N$.

Using this observation on the first order conditions without banking

$$\int_{\tilde{\theta}^N}^{\infty} \left[B_q(Q_t^N, \theta_t) - D_q(Q_t^N) \right] dF(\theta_t, \theta_{t-1}) = 0$$

we can rewrite it as

$$\int_{\tilde{\theta}^N}^{\theta^B} \left[B_q(Q_t^N, \theta_t) - D_q(Q_t^N) \right] dF(\theta_t, \theta_{t-1}) + \int_{\tilde{\theta}^B}^{\infty} \left[B_q(Q_t^N, \theta_t) - D_q(Q_t^N) \right] dF(\theta_t, \theta_{t-1}) = 0.$$

Since $B_{q\theta} > 0$, it holds that the first auxiliary integral is negative, i.e.

$$\int_{\tilde{\theta}^N}^{\tilde{\theta}^B} \left[B_q(Q_t^N, \theta_t) - D_q(Q_t^N) \right] dF(\theta_t, \theta_{t-1}) < 0$$

Thus, the second auxiliary integral has to be positive

$$\int_{\tilde{\theta}^B}^{\infty} \left[B_q(Q_t^N, \theta_t) - D_q(Q_t^N) \right] dF(\theta_t, \theta_{t-1}) > 0.$$

To make the auxiliary integral to match with FOC with banking,

$$\int_{\tilde{\theta}^B}^{\infty} \left[B_q(Q_t^B, \theta_t) - D_q(Q_t^B) \right] dF(\theta_t, \theta_{t-1}) = 0,$$

one can only adjust the cap level. Since $B_{qq} - D_{qq} < 0$, the FOC can be satisfied only if $Q_t^B > Q_t^N$.

The FOC needs to be satisfied with all θ_{t-1} where in some cases the banking motives may not be realized. Therefore, in general $Q_t^B \ge Q_t^N$.

If $\beta < \beta_0$, the banking motives are not realized in any contingency. Therefore, the inter-temporal trading of permits does not have any effect on the policy. As a result, $Q_t^B = Q_t^N$.

B Analytical solution

B.1 Setup

Under strong assumptions, the model can be solved analytically. The assumptions made here are: the benefit and damage functions are quadratic polynomials, the optimal cap is binding without banking with all the realizations of shock θ and the shock θ is IID and follows a continuous, uniform distribution $\theta \sim U[\underline{\theta}, \overline{\theta}]$. Let us denote the marginal damages as

$$D_q(q) = d_0 + d_1 q$$

and marginal benefits

$$B_q(q,\theta) = \theta - b_1 q,$$

where $d_1 > 0$ and $b_1 > 0$. For later use it proves useful to define a dispersion parameter σ through equations $\underline{\theta} = \mathbb{E}\theta - \sigma$ and $\overline{\theta} = \mathbb{E}\theta + \sigma$.

B.2 No banking

Given the assumptions above, the no banking cap is always binding and the current shock realizations conveys no information on about the future shock realizations. Thus, the optimal cap is directly obtained from the first order condition (18)

$$\mathbb{E}\theta - b_1 Q = d_0 + d_1 Q$$

resulting in

$$Q^N = \frac{\mathbb{E}\theta - d_0}{b_1 + d_1},$$

where Q^N denotes the optimal cap level without banking. This result would be valid also if the uniform distribution assumption is relaxed. The condition for always binding cap can be now be written formally as $B_q(Q^N, \underline{\theta}) > 0$, i.e.

$$\sigma < D_q(Q^N),$$

where we have used the first order condition to bring forward the connection between dispersion parameter σ and marginal damages at Q^N .

B.3 Banking

If the discount factor is below the critical value, $\beta \leq \beta_0(Q^N)$, the banking is never performed and the optimal cap with banking is equal to the cap without banking (Proposition 4). The critical cap can be calculated using Definition 3

$$\beta_0(Q^N)D_q(Q^N) = B_q(Q^N, \underline{\theta})$$

leading to

$$\beta_0(Q^N) = \frac{\underline{\theta} - b_1 Q^N}{\mathbb{E}\theta - b_1 Q^N} = 1 - \frac{\mathbb{E}\theta - \underline{\theta}}{\mathbb{E}\theta - b_1 Q^N} = 1 - \frac{\sigma}{D_q(Q^N)}$$

Here we have used the first order condition to interchange the marginal damages and marginal benefits. From the binding cap -assumption it follows that $\beta_0 > 0$.

Let us now concentrate on the case $\beta > \beta_0(Q^N)$ under which positive banking is observed with a positive probability.²³ The optimality condition (18) is with IID shocks

$$\mathbb{E}[B_q(Q,\theta)|\theta > \tilde{\theta}] = D_q(Q)$$

²³Binding cap and positive banking assumption restrict the dispersion parameter: $(1 - \beta)D_q(Q^N) < \sigma < D_q(Q^N)$.

with cut-off shock value

$$\tilde{\theta} = \theta(Q, \beta \mathbb{E}p(Q, \theta')) = \beta \mathbb{E}p(Q, \theta') + bQ,$$
(28)

and price function

$$p(Q,\theta) = \max\{\beta \mathbb{E}p(Q,\theta'), B_q(Q,\theta)\}.$$

The solution of the problem does not involve solving the price function for every argument values as it suffices to solve its expected value. By directly taking expectation from the price function

$$\mathbb{E}p(Q,\theta) = (1 - P(\tilde{\theta}))\beta\mathbb{E}p(Q,\theta) + P(\tilde{\theta})\mathbb{E}[B_q(Q,\theta)|\theta > \tilde{\theta}],$$

where $P(\tilde{\theta})$ stands for probability that the cap is binding, i.e.

$$P(\tilde{\theta}) = \int_{\tilde{\theta}}^{\infty} dF(\theta) = \gamma_0 + \gamma_1 \tilde{\theta}, \qquad (29)$$

with short-hand notation $\gamma_0 := \bar{\theta}/(\bar{\theta}-\underline{\theta})$ and $\gamma_1 := -1/(\bar{\theta}-\underline{\theta})$. Using the first order condition and a short-hand notation $\mu := \mathbb{E}p(Q, \theta)$ we end up with

$$\mu = (1 - P)\beta\mu + P D_q(Q).$$
(30)

Finally we need to calculate the expectation value explicitly

$$\mathbb{E}[B_q(Q,\theta)|\theta > \tilde{\theta}] = \int_{\tilde{\theta}}^{\bar{\theta}} \frac{\theta - b_1 Q}{\bar{\theta} - \tilde{\theta}} d\theta = (\bar{\theta} + \tilde{\theta})/2 - b_1 Q,$$

leading to first order condition

$$(\bar{\theta} + \tilde{\theta})/2 - b_1 Q = d_0 + d_1 Q. \tag{31}$$

Equations (28) - (31) determine the optimal policy which can directly be written as a pair of equations

$$(1 - \beta)\mu = [\gamma_0 + (\beta\mu + b_1Q)\gamma_1](d_0 + d_1Q - \beta\mu)$$

and

$$(\bar{\theta} + \beta \mu + b_1 Q)/2 - b_1 Q = d_0 + d_1 Q$$

Inserting the latter to the first and using the dispersion parameter σ we end up with a quadratic equation for the optimal cap

$$\varphi(Q) := a_2 Q^2 + a_1 Q + a_0 = 0,$$

where

$$a_2 = \beta (d_1 + b_1)^2,$$

$$a_1 = -2\beta (\mathbb{E}\theta + \sigma - d_0)(d_1 + b_1) - (1 - \beta)(2d_1 + b_1)\sigma$$

and

$$a_0 = \beta (\mathbb{E}\theta + \sigma - d_0)^2 - (1 - \beta)(2d_0 - \mathbb{E}\theta - \sigma)\sigma$$

The parabola opens upwards and the minimum of the quadratic polynomial is at emission level

$$Q_{min} = -\frac{a_1}{2a_2} = \frac{\mathbb{E}\theta + \sigma - d_0}{d_1 + b_1} + \frac{1 - \beta}{\beta} \frac{2d_1 + b_1}{2(d_1 + b_1)^2} \sigma,$$

where the first term is the emission level at which the maximum locus of marginal benefits crosses the marginal damages, $B_q(Q, \bar{\theta}) = D_q(Q)$. The second term is positive. Thus, it follows directly that the optimal cap is the smaller of the two roots of the polynomial as the larger one is infeasible solution to the optimization problem. One directly observes that if there is no uncertainty, $\sigma = 0$, the solution reduces to the one without banking. Thus, if there is no stochasticity present, the banking has no role in the optimal policy.

The optimal solution is uniquely determined by level parameter of the marginal damages d_0 and slope parameters of marginal damages and benefits d_1 and b, stochastic marginal benefits level $\mathbb{E}\theta$ and σ as well as discount factor β . The comparative statics can be straightforwardly applied. As the parabola opens upwards and the optimal emission level, Q^* , is the smaller of the two roots, we know that $\varphi_Q(Q^*) < 0$. The effect of dispersion on optimal cap is directly obtained as

$$\frac{\partial Q}{\partial \sigma} = \frac{(2d_1 + (1+\beta)b_1)Q^* - 2\beta(\mathbb{E}\theta + \sigma - d_0) - (1-\beta)(\mathbb{E}\theta + \sigma - 2d_0) - (1-\beta)\sigma}{\varphi_Q(Q^*)}$$

After a short manipulation we end up with

$$\frac{\partial Q}{\partial \sigma} = -\frac{\beta [B_q(Q^*, \bar{\theta}) - \mu] + (1 - \beta)\sigma}{\varphi_Q(Q^*)} > 0,$$

since $B_q(Q, \bar{\theta}) - \mu > 0$, if $\sigma > 0$ and $\beta < 1$. Thus, the optimal cap increases with increasing dispersion. Similarly, by direct calculation we can derive the effect of discount factor

$$\frac{\partial Q}{\partial \beta} = -\frac{\sigma \mu}{\beta \varphi_Q} > 0$$

In both cases we utilized the fact that the optimality conditions require that the endogenous price floor need to be on a auxiliary line

$$p(q) = 2d_0 - \mathbb{E}\theta - \sigma + (2d_1 + b_1)q_2$$

Especially, at the optimum it is needed that $\beta \mu = \underline{p}(Q^*)$. In the comparative statics above, the function $p(Q^*)$ appears in the numerator in both cases.

C Iteration scheme

When the shocks, θ_t , are IID, the current shock realization does not convey information on the future shock realizations. Therefore, the optimal policy does not have any information what to use when assessing the next period market outcomes. As a result, the optimal cap level is a single number, constant over time. However, when the shocks are persistent, the regulator can use the current shock as a signal for the next period outcome. The optimal policy becomes a non-trivial function of current shock, which complicates the optimization of the emission cap. We solve the problem by converting the the continuous shock model into a discrete one and apply an iteration scheme. At period t - 1, the aim is to set an optimal cap for the period t, i.e. $Q_t = Q(\theta_{t-1})$. With banking, the optimal cap level depends on the price expectations by the speculative agents. Since the agents act at period t, the interesting price expectation is $\mathbb{E}_t p_{t+1}$. Correspondingly, these price expectations depend on the cap policy for the next period Q_{t+1} . In a discrete shock case the function of interest, $Q(\theta)$, is transformed into vector \mathbf{Q} where $Q_i = Q(\theta_i)$ for all $i \in \{1, 2, ..., n\}$. Similarly, the expected permit prices form a vector $\boldsymbol{\pi}$ with

$$\pi_i = \mathbb{E}[p(Q_i, \theta_j)|\theta_i] = \sum_j P_{ij} p_{ij},$$

where P_{ij} is the transition probability form θ_i to θ_j and p_{ij} is the realized permit price at θ_j when previous shock realization has been θ_i . The earlier history has no effect on the price realization as the optimal policy truncates the history by nullifying the effect of previous banking decisions.

Iteration scheme for the cap and expected price level pair (Q, π) consists of four steps:

- 1. Choose initial vector π^0 . Set s = 1.
- 2. Find Q^s by solving

$$\max_{\boldsymbol{Q}^s} \sum_{i,j} P_{ij} [B(q_{ij}, \theta_j) - D(q_{ij})],$$

subject to for all i, j:

$$q_{ij} \ge 0,$$

$$p_{ij} = B_q(q_{ij}, \theta_j),$$

$$p_{ij} \ge 0, \qquad Q_i^s - q_{ij} - b_{ij} \ge 0 \quad \text{and} \quad (Q_i^s - q_{ij} - b_{ij})p_{ij} = 0,$$

$$b_{ij} \ge 0, \qquad p_{ij} - \beta \pi_j^{s-1} \ge 0 \quad \text{and} \quad (p_{ij} - \beta \pi_j^{s-1})b_{ij} = 0,$$

3. Calculate π^s

$$\pi_i^s = \sum_j P_{ij} p_{ij}.$$

4. Increase s by one and repeat steps 2. and 3. until required accuracy is reached.

In step 2. we have performed joint maximization for all the contingencies. The optimization can be performed individually cap-levels Q_i , for all contingencies $i \in \{1, 2, ..., n\}$. The equality of the approaches stems from the fact the individual optimization problems are independent of each other. The joint optimization problem grows in size as n^2 . Our GAMS implementation consisted of 111 grid points and resulted in over 50 000 explicit constraints. The gain from joint maximization is that there is no need for loop of separate optimization problems.

The choice of initial expected price vector π^0 is a delicate matter. The initial value has to be such that the optimization problem is still feasible. In practice, this means that the market outcome constraints need to be satisfied. A way to calculate an initial value is to use equation $\pi_i^0 = \sum_j P_{ij} p_{ij}$ with some good initial value for prices p_{ij} .

We apply the constraints resulting from the decisions of market agents through the market clearing and first order conditions. These constraints determine the emission outcome function $\hat{q}(Q_i^s, \theta_j)$ as defined in Lemma 1. In step 4. it is useful to note that using the definition of price permit function of Lemma 1, the iteration step can be formulated as

$$\pi_i^s = \sum_j P_{ij} \max\{\beta \pi_j^{s-1}, B_q(Q_i^s, \theta_j)\}.$$

The mapping can be shown to be a contraction mapping (with bounded B_q) and, thus, the expected prices will converge to a unique, policy dependent, fixed point. However, the numerical tests suggest that the optimization step leaves some numerical disturbances to the solution with the standard tolerance levels. These are caused by the discrete shock approximation. However, the damping of iteration seems to help in the convergence (Judd, 1998, p. 558).

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses: http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1 http://papers.ssrn.com/sol3/JELJOUR_Results.cfm?form_name=journalbrowse&journal_id=266659 http://ideas.repec.org/s/fem/femwpa.html http://www.econis.eu/LNG=EN/FAM?PPN=505954494

http://ageconsearch.umn.edu/handle/35978

http://www.bepress.com/feem/

NOTE DI LAVORO PUBLISHED IN 2015

ERM	1.2015	Elena Verdolini, Laura Diaz Anadon, Jiaqi Lu and Gregory F. Nemet: <u>The Effects of Expert Selection</u> ,
CCCD	0.0045	Elicitation Design, and R&D Assumptions on Experts Estimates of the Future Costs of Photovoltaics
CCSD	2.2015	James Lennox and Ramiro Parrado: <u>Capital-embodied Technologies in CGE Models</u>
CCSD	3.2015	Claire Gavard and Djamel Kirat: <u>Flexibility in the Market for International Carbon Credits and Price</u> Dynamics Difference with European Allowances
CCSD	4,2015	Claire Gavard: Carbon Price and Wind Power Support in Denmark
CCSD	5 2015	Gunnar Luderer, Christoph Bertram, Katherine Calvin, Enrica De Cian and Elmar Kriegler, Implications of
CCDD	0.2010	Weak Neartern Climate Policies on Longtern Mitigation Pathways
CCSD	6 2015	Francisca L André and Luis M. do Castro: Incontinues for Drise Manipulation in Emission Dormit Markets with
	0.2013	Stackolbarg Companying and Luis M. de Castro. Incentives for Trice Manipulation in Emission Fernit Markets with
CCED	7 2015	C. Dispisie Déve Plance and Therees Theles Water Flaus in the Foregroup An Input output Foregroup of the
CCSD	7.2013	C. Diomis Perez blanco and Thomas Thater, water Thows in the Economy. An input-output Framework to
CCCD	9 2015	Assess Water Productivity in the Castle and Leon Keylen (Spain)
CCSD	0.2015	Carlos M. Gomez and C. Dionisio Perez-blanco. <u>Simple Mytris and Basic Matrix about Greening Imgation</u>
CCSD	9.2015	Elorri Igos, Benedetto Rugani, Sameer Rege, Enrico Benetto, Laurent Drouer, Dan Zachary and Tom Haas:
	10 0015	Integrated Environmental Assessment of Future Energy Scenarios Based on Economic Equilibrium Models
ERM	10.2015	Beatriz Martinez and Hipolit Forro: European Natural Gas Seasonal Effects on Futures Hedging
CCSD	11.2015	Inge van den Bijgaart: <u>The Unilateral Implementation of a Sustainable Growth Path with Directed Technical</u>
	10.0015	<u>Change</u>
CCSD	12.2015	Emanuele Massetti, Robert Mendelsonn and Shun Chonabayashi: <u>Using Degree Days to Value Farmland</u>
CCSD	13.2015	Stergios Athanassoglou: <u>Revisiting Worst-case DEA for Composite Indicators</u>
CCSD	14.2015	Francesco Silvestri and Stefano Ghinoi : <u>Municipal Waste Selection and Disposal: Evidences from Lombardy</u>
CCSD	15.2015	Loïc Berger: The Impact of Ambiguity Prudence on Insurance and Prevention
CCSD	16.2015	Vladimir Otrachshenko and Francesco Bosello: <u>Identifying the Link Between Coastal Tourism and Marine</u>
		Ecosystems in the Baltic, North Sea, and Mediterranean Countries
ERM	17.2015	Charles F. Mason, Lucija A. Muehlenbachs and Sheila M. Olmstead: <u>The Economics of Shale Gas</u>
		<u>Development</u>
ERM	18.2015	Anna Alberini and Charles Towe: Information v. Energy Efficiency Incentives: Evidence from Residential
		Electricity Consumption in Maryland
CCSD	19.2015	ZhongXiang Zhang: <u>Crossing the River by Feeling the Stones: The Case of Carbon Trading in China</u>
CCSD	20.2015	Petterson Molina Vale: <u>The Conservation versus Production Trade-off: Does Livestock Intensification</u>
		Increase Deforestation? The Case of the Brazilian Amazon
CCSD	21.2015	Valentina Bosetti, Melanie Heugues and Alessandro Tavoni <u>: Luring Others into Climate Action: Coalition</u>
		Formation Games with Threshold and Spillover Effects
CCSD	22.2015	Francesco Bosello, Elisa Delpiazzo, and Fabio Eboli: <u>Macro-economic Impact Assessment of Future Changes</u>
		in European Marine Ecosystem Services
CCSD	23.2015	Maryse Labriet, Laurent Drouet, Marc Vielle, Richard Loulou, Amit Kanudia and Alain Haurie: Assessment of
		<u>the Effectiveness of Global Climate Policies Using Coupled Bottom-up and Top-down Models</u>
CCSD	24.2015	Wei Jin and ZhongXiang Zhang: <u>On the Mechanism of International Technology Diffusion for Energy</u>
		<u>Technological Progress</u>
CCSD	25.2015	Benjamin Michallet, Giuseppe Lucio Gaeta and François Facchini: <u>Greening Up or Not? The Determinants</u>
		<u>Political Parties' Environmental Concern: An Empirical Analysis Based on European Data (1970-2008)</u>
CCSD	26.2015	Daniel Bodansky, Seth Hoedl, Gilbert Metcalf and Robert Stavins: <u>Facilitating Linkage of Heterogeneous</u>
		Regional, National, and Sub-National Climate Policies Through a Future International Agreement
CCSD	27.2015	Giannis Vardas and Anastasios Xepapadeas: <u>Time Scale Externalities and the Management of Renewable</u>
		Resources
CCSD	28.2015	Todd D. Gerarden, Richard G. Newell, Robert N. Stavins and Robert C. Stowe: <u>An Assessment of the</u>
		Energy-Efficiency Gap and Its Implications for Climate Change Policy
CCSD	29.2015	Cristina Cattaneo and Emanuele Massetti: <u>Migration and Climate Change in Rural Africa</u>
ERM	30.2015	Simone Tagliapietra: The Future of Renewable Energy in the Mediterranean. Translating Potential into
		Reality
CCSD	31.2015	Jan Siegmeier, Linus Mattauch, Max Franks, David Klenert, Anselm Schultes and Ottmar Edenhofer: <u>A Public</u>
		Finance Perspective on Climate Policy: Six Interactions That May Enhance Welfare
CCSD	32.2015	Reyer Gerlagh, Inge van den Bijgaart, Hans Nijland and Thomas Michielsen: Fiscal Policy and CO2 Emissions
		of New Passenger Cars in the EU
CCSD	33.2015	Marie-Laure Nauleau, Louis-Gaëtan Giraudet and Philippe Quirion: <u>Energy Efficiency Policy with Price-</u>
		quality Discrimination

CCSD	34.2015	Eftichios S. Sartzetakis, Anastasios Xepapadeas and Athanasios Yannacopoulos: <u>Regulating the</u>
		Environmental Consequences of Preferences for Social Status within an Evolutionary Framework
CCSD	35.2015	Todd D. Gerarden, Richard G. Newell and Robert N. Stavins: <u>Assessing the Energy-efficiency Gap</u>
CCSD	36.2015	Lorenza Campagnolo and Fabio Eboli: <u>Implications of the 2030 EU Resource Efficiency Target on</u>
		Sustainable Development
CCSD	37.2015	Max Franks, Ottmar Edenhofer and Kai Lessmann: Why Finance Ministers Favor Carbon Taxes, Even if They
		Do not Take Climate Change into Account
CCSD	38.2015	ZhongXiang Zhang: Carbon Emissions Trading in China: The Evolution from Pilots to a Nationwide Scheme
CCSD	39.2015	David García-León: Weather and Income: Lessons from the Main European Regions
CCSD	40.2015	Jaroslav Mysiak and C. D. Pérez-Blanco: <u>Partnerships for Affordable and Equitable Disaster Insurance</u>
CCSD	41.2015	S. Surminski, J.C.J.H. Aerts, W.J.W. Botzen, P. Hudson, J. Mysiak and C. D. Pérez-Blanco: <u>Reflections on the</u>
		<u>Current Debate on How to Link Flood Insurance and Disaster Risk Reduction in the European Union</u>
CCSD	42.2015	Erin Baker, Olaitan Olaleye and Lara Aleluia Reis: <u>Decision Frameworks and the Investment in R&D</u>
CCSD	43.2015	C. D. Pérez-Blanco and C. M. Gómez: <u>Revealing the Willingness to Pay for Income Insurance in Agriculture</u>
CCSD	44.2015	Banchongsan Charoensook: On the Interaction between Player Heterogeneity and Partner Heterogeneity in
		Two-way Flow Strict Nash Networks
CCSD	45.2015	Erin Baker, Valentina Bosetti, Laura Diaz Anadon, Max Henrion and Lara Aleluia Reis: <u>Future Costs of Key</u>
		Low-Carbon Energy Technologies: Harmonization and Aggregation of Energy Technology Expert Elicitation
		Data
CCSD	46.2015	Sushanta Kumar Mahapatra and Keshab Chandra Ratha: Sovereign States and Surging Water: Brahmaputra
		River between China and India
CCSD	47.2015	Thomas Longden: CO ₂ Intensity and the Importance of Country Level Differences: An Analysis of the
		Relationship Between per Capita Emissions and Population Density
CCSD	48.2015	Jussi Lintunen and Olli-Pekka Kuusela: Optimal Management of Markets for Bankable Emission Permits